

GOPEN ACCESS

Citation: Pawłowski KP, Czubak W, Zmyślona J, Sadowski A (2021) Overinvestment in selected Central and Eastern European countries: Production and economic effects. PLoS ONE 16(5): e0251394. https://doi.org/10.1371/journal. pone.0251394

Editor: László VASA, Szechenyi Istvan University, HUNGARY

Received: November 19, 2020

Accepted: April 25, 2021

Published: May 7, 2021

Copyright: © 2021 Pawłowski et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because of the Agreement between DG AGRI-C.3. of the European Commission and Poznań University of Life Sciences in accordance with the provisions of Regulation (EC) No 45/2001 (articles 7,8,9). The agreement obliges the University's authorised persons (i.e.: dr hab. Prof. Wawrzyniec Czubak and MSc Eng. Krzysztof Piotr Pawłowski) to undertakes strictly to observe the rules on non-disclosure of data and the secrecy of statistics; in particular not to communicate data received from DG AGRI C-3 to third parties and not **RESEARCH ARTICLE**

Overinvestment in selected Central and Eastern European countries: Production and economic effects

Krzysztof Piotr Pawłowski **, Wawrzyniec Czubak*, Jagoda Zmyślona*, Arkadiusz Sadowski *

Department of Economics and Economic Policy in Agribusiness, Faculty of Economics, Poznan University of Life Sciences, Poznan, Poland

• These authors contributed equally to this work.

* krzysztof.pawlowski@up.poznan.pl

Abstract

Farms need to invest in order to earn incomes and maintain their competitive edge. However, the scale and extent of investments must be aligned with resources of other productive inputs, primarily including land, because otherwise there is risk of overinvestment. Since 2004, Central and Eastern European countries have been provided with access to investment support programs; these are non-repayable aid funds which can potentially lead to overinvestment issues. Therefore, this paper attempts to answer the question on the scale of overinvestment in the countries covered. This is all the more important since that problem has rarely been addressed in economic and agricultural research. The study presented in this paper is unique in that the research tasks are based on unpublished Farm Accountancy Data Network (FADN) microdata for 5839 selected Central and Eastern European farms provided by the European Commission's Directorate-General for Agriculture and Rural Development (DG AGRI). Based on variables relating to the amount of productive inputs and production volumes, the authors developed their own typology of farms which includes the following categories: optimum investment levels (the growth rate of labor productivity is greater than growth in the assets-to-land ratio); relative overinvestment (while labor productivity grows, it does so at a slower rate than the assets-to-land ratio); absolute overinvestment (labor productivity declines while the assets-to-land ratio grows); underinvestment (decline in both labor productivity and the assets-to-land ratio). The authors demonstrated that members of the 'absolute overinvestment' group made flagrant mistakes in investment planning and implementation, whereas members of the 'relative overinvestment' group did record an improvement in labor productivity which ultimately can be considered a positive outcome. Underinvested farms were a very small minority in each country. In addition to filling a gap in the methodology for determining agricultural overinvestment, this paper also indicates the scale of that issue in Central and Eastern European countries. This study may be of importance both to farms (as guidelines for investment planning) and to agricultural policymakers who develop investment support programs.

to publish a result when it is based on less than 15 observations (holdings). Data are available from the European Commission, DG Agriculture and Rural Development, Unit C.3 Farm Economics (contact via <u>AGRI-C3@ec.europa.eu</u>) for researchers who meet the criteria for access to confidential data.

Funding: K. P. P.: This research was funded by National Science Centre, Poland, grant number UMO-2018/31/N/HS4/03052 (<u>https://projekty.ncn.</u> gov.pl/index.php?projekt_id=430283).

Competing interests: The authors have declared that no competing interests exist.

Introduction

Due to its specific production characteristics related to macro- and microeconomic conditions [1], agriculture is believed to be the main determinant of self-sufficiency in food, protection of natural resources (with main focus on land), rural development and socio-cultural benefits [2]. It also is a part of landscape; for instance, in the European Union (EU), agriculture is the main user of land which accounts for over 47% of total territory [3]. In 2018, the world had a population of 7.6 billion. That number is forecasted to grow, with the greatest increase expected to be recorded in Asian regions such as China, India and Pakistan [4]. Although the Malthusian theory is not supported by how the socioeconomic system evolves, it is worth being referenced to because of its importance in the development of economic theory. It is based on the thesis that population grows exponentially while food production grows arithmetically [5, 6]. Research that identifies links between food, land and population continues to have an effect on how agricultural policy is formulated by international development decision-makers [7, 8]. Also, the considerable increase in the future demand for food will require agricultural production to be intensified in a sustainable way, so that environment may be protected and climatic impacts may be minimized when fulfilling economic functions [9, 10]. Unfortunately, climate change often has a knock-on effect and translates into a number of environmental, agricultural and social consequences while making these matters interdependent [11-15]. Hence, there is need for sustainable development of the agricultural sector in a way to ensure environmental security. Nevertheless, as regards agriculture, human progress means a gradual substitution of labor by capital [16], ultimately leading to growth in farm size, reduction in farm numbers, and less people being employed in agricultural production. These phenomena take place provided that capital is supplied both at sector and farm level as part of investment processes. Investments are of vital importance not only to the continued existence of single operators, for instance due to the effect they have on incomes [17], but also to food security which, in today's world, may be only be ensured through the use of state-of-the-art technological and organizational solutions. Viewed in the long run, food security requires that environmental sustainability and production growth be ensured at the same time [18]. Hence, it is essential to modernize agriculture which generally means attaining progress through improvements in productive, organizational, technical and technological inputs [19, 20]. Furthermore, agricultural progress has an effect on the economy [21].

Numerous studies point to need for agricultural investments because an improvement in agricultural performance is, for obvious reasons, the basis for an increase in food production volumes [22]. The extent to which humans intervene in the food production process to ensure food security (which includes enough food for everyone, access to nutritious food and providing a food guarantee) [23, 24] is usually measured with the amount of investment expenditure [25, 26]. However, no country managed to alleviate poverty and contribute to improvements in food security based on agriculture alone. Nevertheless, investments in agricultural assets provide a framework for reducing poverty in the long run, increase land productivity and reinforce environmental protection measures [27, 28]. Another essential aspect is institutional and industrial development [22] which ultimately also has an impact on the increase in and rationality of agricultural investments.

Nevertheless, one should be aware of the threats involved in the investment process which takes on a particular dimension in the agriculture. A frequent problem is that the importance of investments is underestimated. This is due to the particularities of agriculture which requires an intense, though relatively short, use of fixed assets because of production seasonality. Investments can be economically unviable which means the economic operator suffers a loss in the long term. When attempting to estimate investment viability, it can be noted that

excessive investments (which primarily mean a mismatch between investment level and production scale) ultimately result in operational dysfunctions, including overinvestment. That process is manifested by a "higher capital-to-land ratio" [29], eventually causing a reduction in production efficiency. When considering that problem on an economy-wide or a per-sector basis, it can be concluded that overinvestment results in excess production [30].

In order to identify the extent of agricultural overinvestment, focus must be placed on inefficiencies which occur when the assets-to-labor ratio grows while labor productivity declines [31]. Therefore, this paper addressed the problem of agricultural overinvestment defined as a condition where long-term investments are excessively high compared to the production potential (mainly including land resources) and ultimately become economically unviable [32]. Another method of identifying overinvestment in a publicly subsidized agriculture regime is by calculating the rate of return on public investments in agriculture [33]. The opposite situation (underinvestment) may also occur; this was the case in China when the rates of return on capital investments in agricultural production were much higher than in urban sectors which could be indicative of agricultural underinvestment [34]. As this paper focuses on a microeconomic approach to overinvestment, it examined Central and Eastern European farms in terms of their production and economic performance. The pertinence of addressing this topic is justified because overinvestment in the agricultural sector was observed to affect the use efficiency of resources and to result in a high risk of environmental pollution [35]. Furthermore, no such research has yet been carried out. Although a similar study was undertaken in China based on technical efficiency calculations, it was oriented on production only, and did not take account of the degree of labor productivity [34]. This paper focuses on results underpinned by microdata for selected Central and Eastern European (CEE) countries.

Central and Eastern European countries share a common economic history spanning over the last several decades. From the end of World War 2 until late 1980s, they lived under real socialism, a system which can be briefly characterized as the nationalization of a large part of productive inputs and a centrally planned economy. The fall of that system was accompanied by a crisis which ultimately confirmed its economic inefficiency. However, the economic history of agriculture slightly differs between Central and Eastern European countries [36]. In Poland and Slovenia (until early 1990s, it was part of former Yugoslavia, and was therefore covered by all systemic solutions of that country), private ownership of (family-run) farms was essentially maintained; other countries were dominated by large, nationalized (state-owned or cooperative) holdings. Today, these historical events are reflected in the agrarian structure which differs across countries. For instance, large agricultural holdings play a major role in Czech Republic and Slovakia, whereas other countries are dominated by small family farms while recording a quite considerable share of large enterprises which usually are the former national holdings that underwent a transformation process. Such large operators play the smallest role in Poland and Slovenia [37]. Irrespective of differences referred to above, the crisis of the socialist system resulted in agricultural underinvestment in each of these countries. The farms themselves lacked funds to finance modernization processes and could not rely on state aid. After the fall of real socialism, there were little opportunities to change this situation, including because of the financial deficiency of the restructuring economies. Nevertheless, a number of measures were taken according to the capabilities of each country. In mid-1990s, Poland established a soft loan system primarily intended to finance farm modernization [20, 38]. Central and Eastern European farms saw a great improvement in their upgrading and catching-up opportunities when the respective countries joined the European Union in 2004 and 2007. This is when they became eligible for direct payment schemes and dedicated rural development programs. The latter are of particular importance because they are intentionally largely focused on supporting farm investments. The partially non-repayable form of aid

contributed to a considerable improvement in the situation of many beneficiaries [39], especially as they often had a lot of catching up to do. Nevertheless, the capacity to implement essential projects at a relatively low cost encouraged the farmers to invest excessive amounts of funds in relation to other productive resources owned, especially land. This was the case despite the existence of formal limits for access to investment support programs (usually referred to as the minimum physical or economic size) and inspection procedures being implemented by competent paying agencies. Obviously, public support is not a necessary condition for overinvestment. However, the ability to access additional funds increases that risk, especially because it alleviates some economic restrictions that should have been taken into account by both the investor and the paying agency. Overinvestment can coexist with underinvestment, a phenomenon which has less to do with the use of funds under the Common Agricultural Policy. It is manifested in a situation where essential projects are either not implemented at all or are implemented to such a small extent that they do not prevent the value of assets from decreasing. This may have diverse reasons, including: the farmer not having a successor; the competitive potential being viewed as insufficient by the manager; or the deficiencies in strategic management. In the context of CAP instruments, this may also include the thresholds of access to aid.

The purpose of the work is to determine the scale of the overinvestment in agriculture in the countries of Central and Eastern Europe, which since 2004 have been competing on the single European market and may benefit from pro-investment programs supporting the European Union. Based on the assumption of the subordinate role of investment activity in relation to the operational one, it attempted to answer the questions about how the size of the investment affects the relations between production factors and economic results of farms.

The paper is organized as follows: Section 2 presents pertinence of the topic. Section 3 describes the unique FADN source data and the research methods employed. Section 4 presents and discusses the key findings. Finally, Section 5 presents the summary, conclusions, relevant political implications and guidelines for further research.

Materials and methods

Unpublished FADN (Farm Accountancy Data Network) microdata provided by the European Commission's DG AGRI was used as the source material. The European Farm Accountancy Data Network was established in European Economic Community countries upon initiating the implementation of CAP mechanisms. Note that only commercial farms are monitored by the system. The study presented in this paper is unique in that the research tasks are based on unpublished microdata of selected Central and Eastern European farms. The microeconomic nature of this data makes it also possible to carry out dynamic analyses [40]. Formal guidelines for working with extremely sensitive data are highly restrictive, and therefore this paper only presents aggregated results for no less than 15 farms. The research covered selected Central and Eastern European countries: Bulgaria, Estonia, Hungary, Latvia, Poland and Slovenia. The study period was 2004–2015. The initial year marks the first enlargement of the EU with CEE countries while the last year corresponds to the most recent data from FADN resources. Of all the farms, this study selected only those or which continuous FADN records were available throughout the analysis period (2004–2015 and 2007–2015 for Bulgaria). The number of farms selected in each country is shown in Table 1.

This paper relies on the authors' own classification method which assumes that increasing the value of farm assets through investments is a reasonable thing to do if it results in a proportional growth in labor productivity. Therefore, overinvestment is defined as a situation where:

Bulgaria	Estonia	Hungary	Latvia	Poland	Slovenia
424	241	855	264	3964	82
		-	-	-	·

Table 1. Number of farms covered by this study (by country).

Source: own calculations based on unpublished EU-FADN-DG AGRI data.

https://doi.org/10.1371/journal.pone.0251394.t001

- The increase in the value of assets results in a decline in labor productivity, which may be due to high maintenance costs of particular assets (e.g. depreciation). The above is defined as absolute overinvestment.
- Labor productivity grows at a lower rate than the value of assets. This is referred to as relative overinvestment.

Changes in labor productivity were calculated for each farm. Labor productivity was defined as net value added (gross value added less depreciation) less operating and investment subsidies per FTE. Net value added was used (rather than family farm income) because of the need to eliminate the costs of external inputs (paid labor, rents, interest charged on credits) from the calculation in order to unify the economic performance figures of farms which rely on both their own and external productive inputs. The subsidies were removed from the calculation because public aid cannot be regarded as a metric of labor productivity in the economic sense. This can be assumed even if access to certain subsidies involves (at least formally) the need to perform specific actions, such as meeting the cross compliance or greening requirements in the case of payments. However, these actions refer to the production of public goods, and therefore do not have a direct impact on economic performance recorded in the market. Average values were calculated for the first and last three years of the study period in order to determine the changes. This was used as a basis for calculating the index for changes as per the formula below:

$$LP_{t} = \frac{\sum_{t}^{t+2} \left(\frac{SE410 - SE360 - SE406 - SE605}{SE010} \right)}{3}$$
(1)

$$\Delta LP = \left(\frac{LP_{2013-2015} - LP_{2004-2006}}{LP_{2004-2006}}\right) * 100\%$$
⁽²⁾

where:

LP: labor productivity

SE410: gross farm income

SE360: depreciation

SE406: subsidies on investments

SE605: total subsidies (excluding on investments)

SE010: total labor input (AWU)

Changes in the assets-to-labor ratio were calculated as the next step. The value of fixed assets less the value of land per FTE was used as a metric of the increase in the assets-to-labor ratio. The value of land includes agricultural land, land improvement machinery, permanent crops, quotas and other rights attached (including purchasing costs) and forest land. Production quotas (and other rights attached) received free of charge are not appraised in the balance sheet (only the sales thereof is recorded). The rationale behind the above approach is that over-investment is a problem which ultimately leads to a mismatch between the farm area and the

extent of investments in machinery and buildings. Average values were calculated for the first and last three years of the study period, just like in the case of labor productivity. The index for changes was determined next:

$$ALR_{t} = \frac{\sum_{t}^{t+2} (\frac{SE441 - SE446}{SE010})}{3}$$
(3)

$$\Delta ALR = \left(\frac{ALR_{2013-2015} - ALR_{2004-2006}}{ALR_{2004-2006}}\right) * 100\% \tag{4}$$

where:

ALR: assets-to-labor ratio

SE441: total fixed assets

SE446: land, permanent crops and quotas

SE010: total labor input

As a next step, each farm was attributed to a group based on the overinvestment criterion:

i. Absolute overinvestment; this is the case for farms where labor productivity drops while the assets-to-labor ratio grows:

$$\Delta LP < 0 \quad \wedge \quad \Delta ALR > 0 \tag{5}$$

ii. Relative overinvestment; this is the case for farms where both labor productivity and the assets-to-labor ratio are on an increase but the increase in the assets-to-labor ratio is smaller than the increase in labor productivity:

$$\Delta LP > 0 \quad \land \quad \Delta ALR > 0 \quad \land \quad \Delta LP < \Delta ALR \tag{6}$$

iii. Underinvestment; this is the case for farms where both labor productivity and the assetsto-labor ratio are on a decline.

$$\Delta LP < 0 \quad \wedge \quad \Delta ALR < 0 \tag{7}$$

iv. Optimum investments; this is the case for farms where both labor productivity and the assets-to-labor ratio are on an increase, and labor productivity grows faster than the assets-to-labor ratio:

$$\Delta LP > 0 \quad \land \quad \Delta ALR > 0 \quad \land \quad \Delta LP > \Delta ALR \tag{8}$$

Eight indicators were calculated for each group and for each country separately to identify the reasons for overinvestment. Just like in the case of classification ratios, average values were calculated for the first and last three years of the study period for each indicator. 1. Changes in land resources (ha per farm)

$$I_{1} = \frac{\sum_{t=1}^{n} SE025}{\frac{1}{n}}$$
(9)

where: *SE025*: total utilized agricultural area *n*: number of farms covered by this study (by country)

2. Changes in labor resources (AWU per farm)

$$I_{2} = \frac{\sum_{t=1}^{n} SE010}{\frac{1}{n}}$$
(10)

where: *SE010*: total labor input *n*: number of farms covered by this study (by country)

3. Changes in capital resources (EUR per farm)

$$I_{3} = \frac{\sum_{t=1}^{t+2} (\frac{\sum_{i=1}^{n} (SE441 - SE446)}{n})}{3}$$
(11)

where: *SE441*: total fixed assets *SE446*: land, permanent crops and quotas *n*: number of farms covered by this study (by country)

4. Changes in fixed asset resources in relation to land resources (EUR/ha)

$$I_4 = \frac{\sum_{i=1}^{n} (SE441 - SE446)}{3}$$
(12)

where: *SE441*: total fixed assets *SE446*: land, permanent crops and quotas *SE025*: total utilized agricultural area *n*: number of farms covered by this study (by country) 5. Changes in fixed asset resources in relation to labor resources (EUR/AWU)

$$I_{5} = \frac{\sum_{i=1}^{n} (SE441 - SE446)}{\sum_{i=1}^{n} SE010}$$
(13)

where:SE441: total fixed assetsSE446: land, permanent crops and quotasSE010: total labor input*n*: number of farms covered by this study (by country)

6. Changes in farm productivity (EUR per farm)

$$I_{6} = \frac{\sum_{t=1}^{t+2} (\frac{\sum_{i=1}^{n} SE131}{n})}{3}$$
(14)

where:

SE131: total output

n: number of farms covered by this study (by country)

7. Changes in cost intensity at farm level (EUR per farm)

$$I_{7} = \frac{\sum_{t=1}^{t+2} (\sum_{i=1}^{t-1} SE270)}{3}$$
(15)

where:

SE270: total inputs

n: number of farms covered by this study (by country)

8. Changes in Net Value Added (without taking account of operating and investment subsidies (EUR per farm)

$$I_8 = \frac{\sum_{t=1}^{t+2} (\frac{\sum_{i=1}^{n} (SE410 - SE360 - SE406 - SE605)}{n})}{3}$$
(16)

where: SE410: gross farm income SE360: depreciation SE406: subsidies on investments SE605: total subsidies (excluding on investments) n: number of farms covered by this study (by country) The index for changes was defined as follows for each of the indicators listed above:

$$\Delta I = I_{2013-2015} - I_{2004-2006} \tag{17}$$

Results

The extent of overinvestment and underinvestment differs across Central and Eastern European countries (Fig 1). The relatively greatest number of farms at optimum investment levels can be found in Latvia and the smallest in Estonia (which also is home to the greatest number of farms affected by absolute overinvestment). Relative overinvestment and underinvestment represent a small proportion of cases in each of the countries covered by this study. The main reason for the latter is that after joining the EU, Central and Eastern European countries launched their support programs for farm modernization which essentially consisted in cofinancing the investments on a non-repayable basis. Usually, these forms of support were available to a wide group of beneficiaries (the minimum economic size thresholds were set at a relatively low level) which made overinvestment more likely than underinvestment. Moreover, in accordance with the general requirements for co-financing with Union funds, support programs offered only a partial refinancing of investment expenditure. Hence, it was up to farm managers to assess their competitive position and the likelihood of a return on their investment. As a consequence, when having a relatively small production potential, they intentionally decided not to invest.

Land resources owned by a farm are largely decisive for its competitive position and capacity to generate incomes. Both absolute and relative values are important, especially including how they compare to other farms in the country or region concerned. For reasons referred to above (i.e. a nationalized economy in the socialist era and the dominant privatization path), the majority of farms covered by the FADN (i.e. farms which jointly contribute 90% to standard output) are large farms with an average area above 100 ha (Table 2) in most countries surveyed, the largest being found in Bulgaria and Hungary. The average farm area is definitely smaller in Poland and Slovenia, i.e. countries whose economic history followed a slightly different path when it comes to these aspects. In each case, underinvested farms are the ones with the smallest area; this is true for both periods (2007–2009 and 2013–2015). Also, all countries in this group, except for Slovenia, saw a decline in the average farm area (the sharpest being





Farms	Bulgaria			Estonia			Hungary			Latvia			Poland			Slovenia		
	$\bar{I}_{1_{2007-2009}}$	$\bar{I}_{1_{2013-2015}}$	ΔI_1	$\bar{I}_{1_{2004-2006}}$	$\bar{I}_{1_{2013-2015}}$	ΔI_1	$\bar{I}_{1_{2007-2009}}$	$\bar{I}_{1_{2013-2015}}$	ΔI_1									
Absolute overinvestment	333.5	347.0	13.5	168.7	190.4	21.7	336.6	299.9	-36.7	183.5	226.9	43.4	36.5	43.9	7.4	22.5	23.2	0.7
Relative overinvestment	583.5	545.0	-38.5	219.1	280.6	61.5	258.6	280.3	21.7	430.8	459.5	28.7	35.9	49.9	14.0	14.1	16.4	2.3
Underinvestment	209.4	185.3	-24.1	137.6	89.0	-48.6	149.8	121.0	-28.8	64.0	58.3	-5.7	20.4	19.9	-0.5	12.0	12.2	0.2
Optimum	350.1	326.3	-23.8	429.8	473.4	43.6	192.0	194.9	2.9	339.8	408.0	68.2	32.5	37.5	5.0	15.8	17.0	1.2

Table 2. Land resources (ha	per farm) in farms g	ouped by scale of investment in	2007-2009 and 2013-2015 (I8)
-----------------------------	----------------------	---------------------------------	------------------------------

Source: own calculations based on unpublished EU-FADN-DG AGRI data.

https://doi.org/10.1371/journal.pone.0251394.t002

recorded in Estonia). This suggests that managers of relatively small farms failed to examine their competitive advantages (as described above); they gradually withdraw from farming by reducing investments and selling out their land. Note that what matters in this context is the relative size compared to other farms based in the country concerned. Underinvested farms in Bulgaria, Estonia, Hungary and Latvia are much larger than Polish and Slovenian farms at optimum investment levels. In turn, the largest holdings are relatively overinvested farms (Bulgaria, Latvia, Poland) or absolutely overinvested farms (Hungary, Slovenia). Only in Estonia, the largest farms were found in the group at optimum investment levels. Most farms (except for underinvested ones) saw an increase in their average area, with the highest growth rates being usually recorded by relatively overinvested holdings. Interestingly, farms at optimum investment levels witnessed a small increase in acreage, which could mean they mostly embarked on a capital-intensive development path.

In each country, farm labor resources (Table 3) depend both on the average farm area and on the type of activity. In both periods covered by this study, the amounts of labor were by far the largest in Bulgaria which can be explained by the dominance of a labor-intensive horticulture. A similar situation was found in Hungary (especially in the 'absolute overinvestment' group) and in Latvia (in the 'relative overinvestment' group). In other cases, labor resources are at a level typical of individual farms where hired labor is only used to supplement the farming family's own labor resources. Just like in the case of land, the smallest labor resources are found in underinvested farms. Furthermore, underinvested farms are the ones that experienced the greatest decline in labor resources which, once again, can suggest they gradually discontinue their farming activities (the sharpest reduction was recorded in Estonia and Hungary). Nevertheless, a reduction in labor resources was observed in each country and in nearly every group. Bulgarian farms from the 'absolute overinvestment' group were the only ones to report an increase (by 1.1 AWU). In most countries, the sharpest decline was experienced in the 'relative overinvestment' group. The trend towards reducing labor resources is

	•	1	,	0	1 /							× 27						
Farms	Bulgaria			Estonia			Hungary			Latvia			Poland			Slovenia		
	$\bar{I}_{2_{2007-2009}}$	$\bar{I}_{2_{2013-2015}}$	ΔI_2	$\bar{I}_{22004-2006}$	$\bar{I}_{2_{2013-2015}}$	ΔI_2	$\bar{I}_{2_{2004-2006}}$	$\bar{I}_{2_{2013-2015}}$	ΔI_2	$\bar{I}_{2_{2004-2006}}$	$\bar{I}_{2_{2013-2015}}$	ΔI_2	$\bar{I}_{22004-2006}$	$\bar{I}_{2_{2013-2015}}$	ΔI_2	$\bar{I}_{2_{2004-2006}}$	$\bar{I}_{2_{2013-2015}}$	ΔI_2
Absolute overinvestment	10.8	11.9	1.1	2.9	2.3	-0.6	10.6	9.1	-1.5	4.7	4.9	0.2	2.1	2.2	0.1	2.4	1.9	-0.5
Relative overinvestment	17.8	16.1	-1.7	5.4	5.3	-0.1	6.5	6.4	-0.1	15.8	14.2	-1.6	2.2	2.5	0.3	9.3	8.4	-0.9
Underinvestment	14.9	10.3	-4.6	2.4	1.1	-1.3	6.0	4.5	-1.5	2.2	1.5	-0.7	1.8	1.7	-0.1	1.6	1.4	-0.2
Optimum	17.1	13.2	-3.9	9.9	7.3	-2.6	4.5	4.3	-0.2	9.9	8.6	-1.3	2.2	2.1	-0.1	1.8	1.7	-0.1

Table 3. Labor resources (AWU per farm) in farms grouped by scale of investment in 2007-2009 and 2013-2015 (I2).

Source: own calculations based on unpublished EU-FADN-DG AGRI data.

generally typical of today's modern agriculture characterized by the substitution of labor for capital. However, the change in relationships between labor and capital differs on a case-bycase basis and depends on the particularities of each farm. Indeed, the purposes of investing include an improvement in labor productivity which can be achieved without changes (or even with an increase) in employment [41].

The differences in capital resources between countries are similar to what was found with respect to land. In this context, note that Polish and Slovenian farms generally have a lower value of fixed assets. The attribution of farms to corresponding groups largely depended on changes in capital resources, and therefore it is obvious that the greatest increments in that productive input are recorded either in the 'relative overinvestment' or in the 'absolute overinvestment' group (Table 4). In these groups, the growth rate for capital is (often much) greater than the growth rate for land and labor. Capital resources more than doubled in each farm of the 'relative overinvestment' group. Obviously, in each country, underinvested farms have the smallest capital resources which always continue to decline. Once again, this suggests they gradually discontinue their agricultural activity. All other groups experienced an increase, with the smallest growth being recorded by farms at optimum investment levels (which is largely consistent with the metrological assumptions made in this study). The different investment strategies can only be evaluated in a context of economic effects they attain (as discussed in more detail later in this paper). Nevertheless, if account is taken only of the results shown in Table 3 and of the methodological assumptions of this paper, it can be noticed that as regards the 'absolute overinvestment' group, the increase in capital resources (which was usually considerable) was unreasonable as it failed to translate into improvements in labor productivity. Things look slightly different in the 'relative overinvestment' group: the increase in capital resources was so sharp that the increase in labor productivity could not keep pace with it. This means that either some of the investments were useless or such a considerable increase in capital was necessary to attain a specific improvement in productivity. In the latter case, the availability of EU funds would rather provide an opportunity for improvements in economic performance which otherwise would be impossible or difficult to reach. Note that the changes experienced in that group of farms largely reflect the general characteristics of agriculture, including a great demand for, and underutilization of, fixed assets in the form of machinery and buildings (as field machinery is only used in the growing season), and a relatively low production profitability. Farms at optimum investment levels need to be assessed in that very context. The fact alone that labor productivity grew faster than capital is certainly a significant advantage which means they followed a development strategy that makes efficient use of inputs other than only capital. This could include improvements in farming efficiency based on the adoption of state-the-art techniques and technologies; improvements in production organization; integration; adequate marketing strategies; or, last but not least, improvements in human

Farms	Bulgaria			Estonia			Hungary			Latvia			Poland			Slovenia		
	$\bar{I}_{3_{2007-2009}}$	$\bar{I}_{3_{2013-2015}}$	ΔI_3	$\bar{I}_{3_{2004-2006}}$	$\bar{I}_{3_{2013-2015}}$	ΔI_3												
Absolute overinvestment	305.3	580.1	274.8	130.6	299.2	168.6	638.5	1055.4	416.9	169.3	471.0	301.7	118.8	200.9	82.1	116.3	188.9	72.6
Relative overinvestment	458.5	1015.3	556.8	259.3	681.1	421.8	346.3	813.7	467.4	492.2	1758.8	1266.6	120.6	255.6	135.0	137.1	377.9	240.8
Underinvestment	405.1	256.4	-148.7	120.4	82.7	-37.7	362.3	295.9	-66.4	54.8	40.7	-14.1	79.4	61.2	-18.2	71.5	54.5	-17
Optimum	507.2	749.3	242.1	620.5	904.7	284.2	314.3	420.5	106.2	361.0	753.0	392.0	100.0	124.7	24.7	119.1	184.7	65.6

Table 4. Resources of fixed assets other than land (EUR thousand per farm) in farms grouped by scale of investment in 2007–2009 and 2013–2015 (I₃).

Source: own calculations based on unpublished EU-FADN-DG AGRI data.

capital which accompany the investment process. This is all the more possible since, in most countries covered by this study, these are not the largest farms and therefore it would be reasonable for them to rely on more intensive production methods. When it comes to underinvested farms, all of their characteristics (and, most notably, the trends they follow) suggest they gradually discontinue their agricultural operations. The absence of a competitive edge substantiates their decision. However, irrespective of what drives this process, the performance figures for their productive inputs suggest that refraining from investing inevitably leads to a decline in production potential, manifested by a reducing value of capital and land (and labor, too). Therefore, the situation of underinvested farms should rather be evaluated in a context of multipurpose rural development which, however, goes beyond the scope of this paper. However, it is worth discussing the problem of land resources owned by these farms. Professional mobility and the ability to find a non-agricultural job would result in releasing land to the market. As a consequence, land would be acquired by farms which invest and develop, and could improve their economic performance. This primarily means farms at optimum investment levels and members of the 'relative overinvestment' group. In this situation, current economic decisions on the scale of investment become a determinant of structural changes in agricultural inputs (which are anyway necessary in most CEE countries).

Production efficiency is conditioned by factors which include appropriate relationships between productive inputs, such as the assets-to-land and assets-to-labor ratios. In this context, note that both the excessively low and excessively high levels reduce the farm's ability to maximize economic performance. In the former case, this is because capital deficiencies reduce the use intensity of other productive inputs and restrict the production capacity defined as both the quantity and quality of final output. In the latter case, an excessive amount of capital in relation to other productive inputs (in agriculture, this mainly means land) adds to costs while not increasing the production scale.

In most countries covered by this study, the assets-to-land ratio (Table 5) reached the highest levels in the 'absolute overinvestment' group or the 'relative overinvestment' group (more frequently). Also, these are the groups which experienced the greatest extent of changes (the ratio more than doubled in many cases). It is difficult to clearly assess whether such an indepth transformation is justified. In the first case, this could mean that in the base period, the assets-to-land ratio was excessively low in relation to the production potential, and therefore the changes would be justified. However, the second case could mean overinvestment, i.e. a process whereby capital resources are expanded beyond the productive capacity of land. In accordance with the methodological assumptions adopted in this paper, the second scenario is true. However, as already mentioned, it should be assessed differently depending on whether it took place in the 'absolute overinvestment' or the 'relative overinvestment' group. Indeed, the

Farms	Bulgaria			Estonia			Hungary			Latvia			Poland			Slovenia		
	$\bar{I}_{4_{2007-2009}}$	$\bar{I}_{4_{2013-2015}}$	ΔI_4	$\bar{I}_{4_{2004-2006}}$	$\bar{I}_{4_{2013-2015}}$	ΔI_4	$\bar{I}_{_{42004-2006}}$	$\bar{I}_{4_{2013-2015}}$	ΔI_4	$\bar{I}_{_{42004-2006}}$	$\bar{I}_{4_{2013-2015}}$	ΔI_4	$\bar{I}_{42004-2006}$	$\bar{I}_{_{42013-2015}}$	ΔI_4	$\bar{I}_{4_{2004-2006}}$	$\bar{I}_{_{42013-2015}}$	ΔI_4
Absolute overinvestment	484.6	927.1	442.5	551.0	1118.3	567.0	977.4	2037.8	1060.4	483.1	1418.7	935.6	2539.7	3558.3	1018.6	4707.5	6681.7	1974.2
Relative overinvestment	366.5	831.7	465.2	832.4	1720.7	888.3	622.3	1319.2	696.9	567.4	2396.3	1828.9	2621.0	3933.2	1312.2	8817.7	19505.4	10687.7
Underinvestment	935.1	798.4	-136.7	673.2	667.9	-5.3	1370.8	1320.4	-50.4	416.4	358.9	-57.5	3169.1	2274.0	-895.1	5257.8	3436.6	-1821.2
Optimum	825.6	1271.7	446.1	1134.8	1253.8	119.0	920.9	1115.1	194.2	533.4	1158.2	624.8	2402.7	2351.6	-51.1	7074.8	9150.6	2075.8

Table 5. Value of fixed assets other than land, calculated per hectare of agricultural land (EUR/ha), in farms grouped by scale of investment in 2007–2009 and 2013–2015 (I_4).

Source: own calculations based on unpublished EU-FADN-DG AGRI data.

particularities of natural and technical conditions of agricultural production could result in the capital growth rate (and the growth rate of the assets-to-land ratio analyzed in this paper) being much higher than growth in production volumes or improvements in economic performance. Obviously, positive growth of capital accompanied by a decline in labor productivity (as it is the case in the 'absolute overinvestment' group) is a symptom of manifest errors in both investments and operations. An abnormal situation took place in Bulgaria, Hungary and Poland: in the initial period of this analysis, the highest value of the assets-to-land ratio was recorded in underinvested farms. In these (and other) countries, the assets-to-land ratio declined in that group, and so did other parameters. The largest drop was witnessed in Slovenia and Poland. Estonia was the only country where that ratio remained virtually unchanged. However, it is difficult to view that process as a rationalization of relationships between land and capital resources, for instance because the corresponding ratios in that group (except for Bulgaria, Hungary and Poland, as mentioned earlier) were usually at the lowest levels already in the first years of this analysis. Hence, the decline in that ratio rather seems to corroborate the supposed gradual discontinuation of farming activities.

The potential of today's agriculture largely depends on the use of capital and of state-of-theart technologies. Therefore, the higher the assets-to-labor ratio, the greater the competitiveness of a farm. In reality, the inhibiting factor are land resources which either do or do not enable an efficient use of capital owned. This is why in most countries surveyed (Table 6), the highest levels of and the greatest increments in the assets-to-land ratio are found in the 'relative overinvestment' or the 'absolute overinvestment' group. In Estonia and Hungary, these ratios were at the lowest level in both groups in the base period. However, they reached the highest levels in 2013–15 as a consequence of investments. The situation is different in Bulgaria and Slovenia: in both periods, the highest values of the assets-to-labor ratio were recorded in farms at optimum investment levels. However, the changes were relatively minor which could suggest these farms were highly reasonable in their investment activities. Note also that in most countries, underinvested farms reported the lowest values of that ratio; however, it followed a slight upward trend (except for Poland and Slovenia) driven by the reduction in labor force being faster than the reduction in capital rather than by the implementation of investments (Tables <u>3</u> and <u>4</u>).

It is difficult to estimate the optimum relationships between productive inputs as they depend on a number of aspects, e.g. activity type (a more intensive activity intrinsically requires greater amounts of capital), cyclical conditions (price relationships) and natural factors (climate and soil conditions). Furthermore, a major role is played by non-measurable factors specific to each farm, such as skills, knowledge and experience of the manager; risk management; production integration, diversification or specialization; particularities of the local market; and the social and economic environment. Nevertheless, the status of and changes in the relationships between productive inputs can be primarily assessed in the context of production and economic performance of a farm or a group of farms.

Table 0. Value of fixed assets other than failu, calculated per employee (LOR/AVVO), in failing grouped by scale of investment in 2007-2007 and 2013-20	ployee (EUR/AWU), in farms grouped by scale of investment in 2007–2009 and 2013–2015	r employ	 Value of fixed assets other than land, calculated 	Table 6.
---	--	----------	---	----------

Farms	Bulgaria			Estonia			Hungary			Latvia			Poland			Slovenia		
	$\bar{I}_{5_{2007-2009}}$	$\bar{I}_{5_{2013-2015}}$	ΔI_5	$ar{I}_{5_{2004-2006}}$	$\bar{I}_{5_{2013-2015}}$	ΔI_5	$\bar{I}_{5_{2004-2006}}$	$\bar{I}_{5_{2013-2015}}$	ΔI_5	$\bar{I}_{5_{2004-2006}}$	$\bar{I}_{5_{2013-2015}}$	ΔI_5	$\bar{I}_{5_{2004-2006}}$	$\bar{I}_{5_{2013-2015}}$	ΔI_5	$ar{I}_{5_{2004-2006}}$	$\bar{I}_{5_{2013-2015}}$	ΔI_5
Absolute overinvestment	15.0	27.2	12.2	32.1	93.9	61.8	31.2	67.0	35.8	18.8	66.2	47.4	44.5	70.6	26.1	44.3	83.1	38.8
Relative overinvestment	12.0	28.2	16.2	34.1	90.8	56.7	24.7	57.4	32.7	15.5	77.4	61.9	42.2	78.6	36.4	13.3	38.2	24.9
Underinvestment	13.3	14.4	1.1	38.7	54.0	15.3	34.3	35.3	1.0	12.0	14.2	2.2	35.7	26.6	-9.1	38.4	30.9	-7.5
Optimum	16.9	31.5	14.0	49.4	81.5	32.1	39.0	51.1	12.1	18.3	55.1	36.8	35.9	42.5	6.6	61.2	93.1	31.9

Source: own calculations based on unpublished EU-FADN-DG AGRI data.

Production and economic performance figures are the ultimate validation of investment decisions. Indeed, it has to be assumed that investment activities are an auxiliary function to operations because the objective of each enterprise (including farms) is to maximize economic value. In turn, from the social point of view of today's expectations, agriculture requires more and more capital resources and innovative solutions in order to meet the challenge of feeding a growing population, especially as only a small proportion of workforce are employed in agriculture in highly developed countries. However, this general conclusion is not always reflected in farm-level realities. As mentioned earlier, at a micro level, the basic barrier to a reasonable use of capital is the area of utilized land. This is why excessive (unfounded) investments can often lead to stagnation in production and to a deterioration in economic performance. Obviously, the consequences of a lack of investments can be similar or even worse. In the countries covered by this study (Table 7), production grows in all groups except for underinvested farms (only Slovenia recorded a growth of EUR 2,700). This is the consequence of both investment activities and improvements in business conditions after the accession to the EU. In most countries, just like in the case of other ratios, the highest values and the greatest changes were recorded in the 'relative overinvestment' group (which is all the more understandable since these are usually the largest farms). However, the characteristic feature of this group is that capital grows faster than production. A similar situation occurs in the 'absolute overinvestment' group. Abnormal patterns can be found in Hungary where the greatest production volumes are recorded in the 'absolute overinvestment' group, and in Estonia where farms at optimum investment levels report the greatest production volumes.

Once completed, investments can contribute to either a reduction or an increase in costs. The use of more advanced and more efficient machinery and equipment can help making production less energy-intensive, e.g. by reducing the number of runs or by using more fuel-efficient engines. However, the fact alone of having additional fixed assets generates costs such as depreciation, insurance or repair. Moreover, the changes in cost levels should go hand in hand with changes in production levels, ultimately leading to improvements in economic performance. In most cases covered by this study, the 'relative overinvestment' and 'absolute overinvestment' groups (Table 8) reported the largest amounts of and the highest growth rates for total costs, except for Estonia where the highest costs were incurred by farms at optimum investment levels. In all countries, underinvested farms had the smallest costs and the lowest growth rates. Generally, the amount of total costs was on an upward trend. This could result both from the increase in prices of basic productive inputs and from the increased production intensification (leading to a larger scale of production).

They key metrics of viability of investment decisions are the level of and, first of all, the changes in economic performance figures. The performance figures for the base year reflect the resources of productive inputs (primarily land and capital) owned by farms, the skills in

Farms	Bulgaria			Estonia			Hungary			Latvia			Poland			Slovenia		
	$\bar{I}_{6_{2007-2009}}$	$\bar{I}_{6_{2013-2015}}$	ΔI_6	$\bar{I}_{6_{2004-2006}}$	$\bar{I}_{6_{2013-2015}}$	ΔI_6												
Absolute overinvestment	237.7	350.6	112.9	70.3	130.9	60.6	433.8	583.8	150.0	95.8	203.2	107.4	47.2	75.0	27.8	23.3	43.3	20.0
Relative overinvestment	477.6	745.2	267.6	151.1	383.7	232.6	251.0	433.3	182.3	315.4	818.9	503.5	52.4	114.9	62.5	68.3	172.3	104.0
Underinvestment	190.8	175.3	-15.5	48.7	27.6	-21.1	207.3	171.5	-35.8	30.9	25.0	-5.9	26.9	26.4	-0.5	12.9	15.6	2.7
Optimum	325.6	645.9	320.3	274.4	554.5	280.1	159.8	249.7	89.9	191.3	407.0	215.7	37.9	69.2	31.3	25.0	59.1	34.1

Table 7. Production value (EUR thousand per farm) in farms grouped by scale of investment in 2007-2009 and 2013-2015 (I₆).

Source: own calculations based on unpublished EU-FADN-DG AGRI data.

Farms	Bulgaria			Estonia			Hungary			Latvia			Poland			Slovenia		
	$\bar{I}_{7_{2007-2009}}$	$\bar{I}_{7_{2013-2015}}$	ΔI_7	$ar{I}_{7_{2004-2006}}$	$\bar{I}_{7_{2013-2015}}$	ΔI_7	$\bar{I}_{7_{2004-2006}}$	$\bar{I}_{7_{2013-2015}}$	ΔI_7									
Absolute overinvestment	225.1	417.4	192.3	66.6	158.4	91.9	466.9	665.2	198.3	93.1	231.1	138.0	34.2	69.1	34.8	23.5	53.9	30.4
Relative overinvestment	431.0	728.4	297.4	138.2	389.7	251.5	257.1	411.0	153.8	312.1	826.6	514.5	37.3	86.6	49.3	62.7	143.2	80.5
Underinvestment	187.2	223.3	36.1	46.8	38.9	-7.9	225.1	207.5	-17.6	29.6	31.9	2.3	20.9	24.4	3.5	14.0	22.5	8.5
Optimum	386.7	585.0	198.3	290.5	580.5	289.9	187.2	240.6	53.4	216.6	421.5	204.8	31.1	50.7	19.7	24.3	47.7	23.4

	Table 8. Total production costs (E	EUR thousand per farm) in	farms grouped by scale of inve	stment in 2007-2009 and 2013-2015 (I7
--	------------------------------------	---------------------------	--------------------------------	---------------------------------------

Source: own calculations based on unpublished EU-FADN-DG AGRI data.

https://doi.org/10.1371/journal.pone.0251394.t008

managing the production process and finance, and the general condition of the economy. Conversely, the extent of changes indicates the reasonability of the farms' strategic measures. The differences between farm groups in each country are an aspect of particular importance in this context because they indicate the effect of various (mainly endogenous) factors. Moreover, as the farms covered by this study operate in the EU single market and in the geographic vicinity of Europe, it can also be assumed that—irrespective of their home country—all of them were faced with the same general economic conditions (mainly including price conditions).

Economic performance was measured as Net Value Added (NVA) less operating subsidies (mainly including direct payments; payments under the 2nd pillar of the CAP other than investment support; and national support, if any). NVA itself is the difference between production value and production costs less costs of external inputs (rents, interest on loans, and expenditure on hired labor), adjusted by the balance of taxes and operating payments. Hence, it measures the payment for the use of all productive inputs irrespective of who owns them. As a consequence, it can be used in comparing farms irrespective of whether they operate based on their own or external productive inputs. The amount of subsidies was deducted from NVA because these analyses refer to economic outcomes attained by farms as part of their market activities, and are supposed to indicate the actual efficiency of production processes. The objective of external support is to mitigate the agricultural effects of market failures. However, when taken into account in an analysis, subsidies could distort the true picture of actual outcomes.

In the countries surveyed, the highest NVA levels were recorded in the 'relative overinvestment' group. However, the greatest changes were usually witnessed by farms at optimum investment levels (Table 9). In the final period of this analysis (2013–2015), both of these groups had the greatest surplus in each country. Importantly, an increase in NVA was only recorded in that group. In this context, it seems interesting that farms who exhibited a passive attitude towards investments (underinvested farms) as well as those characterized by excessive

		-	-	-			-				-							
Farms	Bulgaria			Estonia			Hungary			Latvia			Poland			Slovenia		
	Ī 82007-2009	$\bar{I}_{82013-2015}$	ΔI_8	$\bar{I}_{82004-2006}$	$\bar{I}_{82013-2015}$	ΔI_8	$\bar{I}_{8_{2004-2006}}$	$\bar{I}_{82013-2015}$	ΔI_8	$\bar{I}_{82004-2006}$	$\bar{I}_{8_{2013-2015}}$	ΔI_8	$\bar{I}_{8_{2004-2006}}$	$\bar{I}_{8_{2013-2015}}$	ΔI_8	$\bar{I}_{82004-2006}$	$\bar{I}_{82013-2015}$	ΔI_8
Absolute overinvestment	64.9	21.4	-43.41	8.8	-12.0	-20.86	77.3	28.6	-48.71	8.7	-8.7	-17.43	15.7	6.7	-8.97	-1.0	-13.3	-12.23
Relative overinvestment	165.8	231.5	65.72	35.2	63.2	28.04	58.0	103.7	45.67	61.8	135.6	73.77	18.3	32.0	13.69	32.0	75.2	43.21
Underinvestment	58.8	8.1	-50.70	3.1	-9.9	-13.00	33.4	5.7	-27.69	2.5	-5.6	-8.10	6.9	2.4	-4.54	-2.3	-7.3	-5.00
Optimum	18.1	194.5	176.45	29.4	82.6	53.24	12.7	58.9	46.16	-0.5	49.5	50.08	9.2	21.7	12.52	-1.4	11.5	12.96

Table 9. Net value added (excluding operating subsidies) (EUR thousand per farm) in farms grouped by scale of investment in 2007-2009 and 2013-2015 (I8).

Source: own calculations based on unpublished EU-FADN-DG AGRI data.

activity (the 'absolute overinvestment' group) ended up with a decline in economic performance. The sole difference between them is that the former have a much greater net value added in the final period. In the case of underinvested farms, this is consistent with the strategy of a gradual discontinuation of farming operations, as described above. However, when it comes to relatively large farms from the 'absolute overinvestment' group, this is a symptom of manifest errors in investing which entail certain (often adverse) changes in operations. Another equally interesting and important conclusion from this study is that the relationships discussed above (i.e. an increase in NVA in farms at optimum levels of investment and in the 'absolute overinvestment' group) are true for all countries, irrespective of their particularities.

Discussion

The particularities of each sector have a considerable effect on investment methods. When comparing agriculture to industrial sectors, for instance, it cannot be ignored that industrial investments are both larger and more frequent. This is especially true for the heavy industry where expenses are focused on the industrial infrastructure (e.g. investment in industrial complexes) [42]. The basic barriers to investment include financial restrictions and imperfections of the loan market [43]. Investments are viewed as the driving force for long-term economic growth [44]. The literature relating to investments increasingly emphasizes that, at least in developed countries, agricultural investments are subsidized (which allows to ensure food security), whereas the implementation of state-of-the-art techniques and technologies makes it possible for agricultural activities to have a smaller environmental and climatic impact. Conversely, the lack of investments often leads to decelerated growth of farm productivity [45, 46]; at the same time, obsolete technologies have a considerable adverse impact on the environment.

For these reasons, the need to make agricultural investments in the context of a major systemic shift (including the emergence of a number of technologies designed to improve labor productivity) should not give rise to controversy. Particular focus should be placed on investments in precision farming or, very soon, in the Internet of Things or the blockchain technology which are consistent with Agriculture 4.0, a concept that may affect the way food is produced, processed, traded and, last but not least, consumed [47–49]. Generally, the importance of technological progress in agriculture is measured based on relationships between resources [50]. This means metrics such as the amount of capital per employee, the amount of capital per hectare of agricultural land and the amount of agricultural land per employee.

Overinvestment is a condition where long-term investments are excessively high compared to the production potential (which, in the agriculture, mostly consists of land resources) and ultimately become economically unviable because the costs incurred to buy and maintain assets were higher than production outcomes. The term "overinvestment" itself is more often used in an industrial or corporate context, and the way it affects agriculture is described rarely (despite being a common occurrence). According to some studies, public funds used in addition to conventional inputs also greatly contributed to an increase in agricultural production [51, 52]. This is a finding from macroeconomic research carried out in China. According to the literature, overinvestment (and the resulting excess production) is increasingly often among the reasons behind massive Chinese exports [53–55]. The situation is different at company or farm level: instead of excess production, overinvestment means a decline in economic performance.

Various types of (industrial, agricultural, service) enterprises get over-invested through an irrational use of assets which is intended to increase company value or have a positive effect on financial performance, but is based on an excessively optimistic evaluation of market

conditions [56–58]. As a consequence, the expected return on investment projects is below the interest rate offered in capital markets [59, 60]. According to another approach, overinvestment takes place when companies excessively invest in financial or tangible assets [58]. This sometimes happens in the case of external financing (e.g. subsidies) which is intended to be used as specifically indicated by the financing authority. Investment financing decisions should be made by the economic operator concerned, because a relationship exists between external financing and investment decisions [61, 62]. The capital market results in extreme financial restrictions with sub-optimal investment strategies as the ultimate consequence [63]. The financial leverage is another aspect related to corporate overinvestment. If external financing is readily available, as it is the case e.g. in monetary expansion periods, debt is no longer a traditional management control tool whereas excessive systemic liquidity results in the emergence of leverage which encourages investment and, hence, exacerbates the problem of overinvestment [64]. Sustainable investments in both the agricultural and food sector are necessary [65], even though they account for a relatively minor proportion of global capital markets [66].

European agriculture is based on plant production, primarily including the cultivation of cereals and industrial crops [67]. Moreover, countries which are able to produce food for both humans and animals focus their production efforts on meat, meat products and dairy [68]. Before joining the European Union (EU), Central and Eastern European Countries were characterized by a much lower profitability of agriculture than "old" member countries. The accession to the EU was expected to result in an increase in productivity accompanied by a reduction in net agricultural employment [69]. Factors which have an impact on ensuring food security include the Common Agricultural Policy (CAP) implemented in the European Union; one of its initial objectives was to increase agricultural production [70–72]. The CAP was created at a time where food expenses accounted for a large proportion of earnings of the population of member states [73]. Productivity was supposed to improve through measures which include the investment impetus [74]. "Investments in fixed assets," a measure available under the 2^{nd} pillar of the CAP, is one of the important instruments in this respect. Its objective is to support farm investments and improve the structure of farming land [75]. Major priorities of the CAP after 2020 will include knowledge and technology investments which enable environmental protection and the existence of low-carbon economies [76]. The grounds for agricultural interventionism result from the conditions of the biological nature of agricultural production [77, 78]. Other aspects that substantiate it include informational imperfection, income-related problems of agriculture and the incompleteness and imperfection of markets related to agriculture [79]. Production periodicity and seasonality and economic fluctuations are the reasons why agricultural investment are less effective than in other sectors. Also, the biological growth and development process makes outputs distant in time from inputs while decelerating the movement of capital [21]. Anyway, agricultural subsidies are a guarantee of food security which means that investments should always drive positive production and economic effects. This is particularly true for investment support. In addition to the basic reason (as mentioned above) which is the society's and government's concern for an adequate supply of high-quality foods, note also the particular way of using most of agricultural equipment. Unlike industrial machinery or vehicles, agricultural equipment is only used during the growing season which is a few months in most countries around the world. Also, the extent of using it is limited by the size of the farm and, as the case may be, by local demand for agricultural services. As a consequence, field equipment is used for a relatively small part of the year, and therefore the investment and operating expenses per hour of work are much higher than in other sectors. Despite these shortcomings, a farm must own high-performance equipment primarily due to general societal reasons (as it enables efficient production of food by a relative

small group of people). However, at a microeconomic level, the farmers lack rationale to purchase it, and thus the need arises for public support.

One of the reason to overinvestment in agriculture is also a lack of trust among farmers. Trust is very important in human relations and cooperation in business relationship, especially in agriculture [80, 81]. The trust in agriculture (often leading to formal integration relationships) has the potential to benefit farmers in gaining access to the innovation and knowledge, increase the productivity, quality and cost reduction. Effective cooperation in agriculture allows, for example, the collective use of machinery, common purchases of production inputs and sales of final products, reduction of transaction costs. Unfortunately, the moral hazardon the machinery sharing cooperations arrangements as for example—is visible and have negative influence on the cooperation [82]. The moral hazard would be less if all investment subsidies were burdened with individual responsibility [83]. The moral hazard is observed on credit markets [84], including preferential loan in agriculture in CEE. The lack of trust is the most important reason why relationship between farmers do not work well [85]. Poor management of agribusiness establishments can lead to moral hazards that can affect investments in agriculture [86]. The research of Kovacs [87] shows that the performance of Hungarian producer organizations in agriculture is not as developed as in the Western European countries and the moral hazard has an impact on the farmers' collaboration. The propensity to overinvest is also implied by the propensity to take risks. Overall, low risk inclination among farmers cause under-investment [88]. Every investment comes with a certain amount of risk [89], especially in agriculture. The investment in agriculture is partially financed by public funds often which lowers the risk score of the investment but don't doesn't reduce it entirely [90].

Conclusions

For Central and Eastern European countries, joining the EU was a great leap forward, including in the agricultural sector. Aid funds allocated to farm investments played an important role in that respect, allowing many operators to catch up with progress (some of them dealing with a modernization backlog dating back to the real socialism era). The investments implemented by some farms were excessive in relation to their production potential. Conversely, other ones gradually discontinued their farming operations as they saw no opportunity for gaining a competitive edge.

This study confirmed that investments have a complementary nature vis-à-vis operations. Indeed, production processes require adequate capital resources, whereas economic performance is what ultimately validates investment decisions. This paper used two essential criteria in assessing these aspects: the absolute increase in economic outcomes, and how it relates to the increase in capital resources. This allowed to identify two types of farms, namely the 'absolute overinvestment' and 'relative overinvestment' groups. The first case is a symptom of manifest errors in investing (and, perhaps, in how operations are conducted), since large amounts of investment expenditure were accompanied by a decline in economic performance. In this context, note that it was not the consequence of unfavorable price relationships because other operators recorded improvements despite dealing with the same market conditions. Things look slightly different in the 'relative overinvestment' group. They saw an improvement in economic performance, which is the goal sought by every economic operator. However, the growth was slower than the increase in capital resources. On the one hand, this could suggest excessive investment took place; but on the other hand, these measures could have been required because of the particularities of production. The second interpretation seems more justified as that group exhibited the highest net value added in all countries surveyed (although in practice both scenarios are likely to have taken place).

When assessing the Union program for investment support on an overall basis, it is necessary to exercise extreme caution in the interpretation because some of the relevant decisions were made otherwise than in the context of availability of aid funds. Nevertheless, support from the EU was significant and must be evaluated. Undoubtedly, farms at optimum investment levels should be viewed as being in a positive situation; in their case, economic performance grew faster than capital. In the context of this discussion (although with some reservations presented above), the activity of the 'relative overinvestment' group should also be considered a positive development. Despite a decline in most ratios, there was a specific rationale behind the measures taken by small and underinvested farms. Dealing in a specific national context and not having a competitive edge, they gradually discontinued their farming activities. Hence, actual and manifest errors were made in the 'absolute overinvestment' group. When assessing the rationality and efficiency of the Union support program, note that the share of this group in the total number of farms ranged from a few to more than ten percent, depending on the country (it went beyond 20% only in Poland). This means that despite being tempted to abuse the non-repayable aid scheme-most farmers did the right thing and acted responsibly, in accordance with their market position. It could be explained by the fact the farmers contributed their own funds and realized that assets generate costs. The formal barriers to access aid programs and the verification of project plans by the granting institutions could have played a certain role, too.

Author Contributions

Conceptualization: Krzysztof Piotr Pawłowski, Wawrzyniec Czubak.

Data curation: Krzysztof Piotr Pawłowski, Wawrzyniec Czubak.

Formal analysis: Krzysztof Piotr Pawłowski, Wawrzyniec Czubak.

Funding acquisition: Krzysztof Piotr Pawłowski.

Investigation: Krzysztof Piotr Pawłowski, Wawrzyniec Czubak.

- Methodology: Krzysztof Piotr Pawłowski, Wawrzyniec Czubak, Jagoda Zmyślona, Arkadiusz Sadowski.
- Project administration: Krzysztof Piotr Pawłowski.

Resources: Wawrzyniec Czubak.

- Writing original draft: Krzysztof Piotr Pawłowski, Wawrzyniec Czubak, Jagoda Zmyślona, Arkadiusz Sadowski.
- Writing review & editing: Krzysztof Piotr Pawłowski, Wawrzyniec Czubak, Jagoda Zmyślona, Arkadiusz Sadowski.

References

- Czubak W., Pawłowski K. P., Wiza P. L., Arion F. (2019). Polish and Romanian dairy farms using EU investment support: a comparative study. *Annals of the Polish Association of Agricultural and Agribusi*ness Economists, XXI(4), 64–75.
- 2. Far S. T., Razaei-Moghaddam K. (2019). Multifunctional agriculture: an approach for entrepreneurship development of agricultural sector. *Journal of Global Entrepreneurship Research*, 9(23).
- Giannakis E., Bruggeman A. (2015). The highly variable economic performance of European agriculture. Land Use Policy, 45, 26–35.
- Tikas G. D., Akhilesh K. B. (2020). Smart Agriculture: A Tango Between Modern IoT-Based Technologies and Traditional Agriculture Techniques [in:] Akhilesh K., (eds) Möller D. (eds) Smart Technologies, 387–393.

- Burger J. R. (2019). Malthus on Population. [in:] Shackelford T., (eds), Weekes-Shackelford V. (eds) Encyclopedia of Evolutionary Psychological Science. Springer, Cham.
- 6. Fischer C. (2019). Avoiding Malthus 2.0: Why Food Pessimism Leads Nowhere. *American Journal of Biomedical Science & Research*, 5, 64–65.
- 7. Fennell S. (2019). Malthus, statistics, and the state of Indian agriculture. *The Historical Journal*, 63(1), 1–27.
- 8. Madsen J. B., Robertson P. E., Ye L. (2019). Malthus Was Right: Explaining a Millennium of Stagnation. *European Economic Review*, 118, 51–68.
- Sanaullah M., Usman M., Wakeel A., Cheema S.A., Ashraf I., Farooq M. (2020). Terrestrial ecosystem functioning affected by agricultural management systems: A review. *Soil & Tillage Research*, 196, 104464.
- Mason-D'Croz D., Sulser T. B., Wiebie K., Rosegrant M. W., Lowder S. K., Nin-Pratt A., et al. (2019). Agricultural invetsments and hunger in Africa modelling potential contributions to SDG2 –Zero Hunger. World Development, 116, 38–53 https://doi.org/10.1016/j.worlddev.2018.12.006 PMID: 30944503
- Khanian M., Marshall N., Zakerhaghighi K., Salimi M., Naghdi A. (2018). Transforming agriculture to climate change in Famenin County, West Iran through a focus on environmental, economic and social factors. Weather and Climate Extremes, 21, 52–64.
- 12. Pauw E. D., Ramasamy S. (2020). Rapid detection of stressed agricultural environments in Africa under climatic change 2000–2050 using agricultural resource indices and a hotspot mapping approach. *Weather and Climate Extremes*, 27, 100211.
- Williges K., Mechler R., Bowyer P., Balkovic J. (2017). Towards an assessment of adaptive capacity of the Europesna agricultural sector to droughts. *Climate Services*, 7, 47–48.
- 14. Peskett L., Grist N., Hedger M., Lennartz-Walker T., Scholz I. (2009). Climate change challenges for EU development co-operation: emerging issues Working Paper, EDC 2020, no. 3.
- 15. Daszkowska E. (2008). Przyczyny i odczuwalne skutki kryzysu żywnościowego pierwszej dekady XXI wieku. Działania stabilizujące sytuację, podejmowane przez Unię Europejską i Bank Światowy (Reasons behind and impacts of the 2000s food crisis. Stabilization measures taken by the European Union and the World Bank). Journal of Agribusiness and Rural Development, 4(10), 17–24.
- Niezgoda D. (2009). Uwarunkowania rentowności gospodarstw rolnych zróżnicowanych pod względem ich wielkości ekonomicznej (Conditions for the profitability of farms of different economic size). Agricultural Sciences Yearbook, G series, 96, 155–165.
- 17. Scaramuzzi S., Belletti G., Biagioni P. (2020). Integrated Supply Chain Projects and multifunctional local development: the creation of a Perfume Valley in Tuscany. *Agric Econ*, 8, 5.
- Garnett T., Appleby M. C., Balmford A., Bateman I. J., Benton T. G., Bloomer P., et al. (2013). Sustainable Intensification in Agriculture: Premises and Policies. *Science Magazine*, 341, 33–34. <u>https://doi.org/10.1126/science.1234485</u> PMID: 23828927
- Whitcraft A. K., Becker-Reshef I., Justice Ch. O., Gifford L., Kavvada A., Jarvis I. (2019). No pixel left behind: Toward integrating Earth Observations for agriculture into the United Nations Sustainable Development Goals framework. *Remote Sensing of Environment*, 235, 111470.
- 20. Sadowski A. (2011). Skutki inwestycji wspieranych kredytem preferencyjnym w polskim sektorze rolnym w świetle wyników badań (Effects of investments supported with soft loans in the Polish agricultural sector: evidence from research). [in:] Projekty inwestycyjne w agrobiznesie a zasady Wspólnej Polityki Rolnej po 2013 roku (Agribusiness investment projects vs. principles of the Common Agricultural Policy after 2013). eds. Czyżewski A., Poczta W., Publishing House of the Poznań University of Economics and Business, Poznań, Poland, 176–179.
- Czubak W. (2013). Rozwój rolnictwa w Polsce z wykorzystaniem wybranych mechanizmów Wspólnej Polityki Rolnej Unii Europejskiej (Agricultural development in Poland based on selected mechanisms of the European Union's Common Agricultural Policy), Publishing House of the Poznań University of Life Sciences, Poznań, Poland.
- 22. Ogundari K. (2014). The Paradigm of Agricultural Efficiency and its Implication on Food Security in Africa: What Does Meta-analysis Reveal? *World Development*, 64, 690–702.
- Estrada-Carmona N., Raneri J. E., Alvarez S., Timler C., Chatterjee S. A., Ditzler L., et al. (2019). A model-based exploration of farm-household livelihood and nutrition indicators to guide nutrition-sensitive agriculture interventions. *Food Security*, 12, 59–81.
- Brooks J. (2014). Policy coherence and food security: The effects of OECD countries agricultural policies. Food Policy, 44, 88–94.
- Baer-Nawrocka A., Sadowski A. (2019). Food security and food self-sufficiency around the world: A typology of countries. *PLOS ONE*, 14(3), e0213448. https://doi.org/10.1371/journal.pone.0213448 PMID: 30845273

- Cassman K. G., Harwood R. R. (1995). The nature of agricultural systems: food security and environmental balance. *Food Policy*, 20(5), 439–454.
- Tomich T.P., Lidder P., Coley M., Gollin D., Meinzen-Dick R., Webb P., et al. (2019). Food and agricultural innovation pathways for prosperity. *Agricultural Systems*, 172, 1–15.
- Andrade J. F., Rattalino Edreira J. I., Farrow A., van Loon M. P., Craufurd P. Q., Rurinda J., et al. (2019). A spatial framework for ex-ante impact assessment of agricultural technologies. *Global Food Security*, 20, 72–81.
- 29. Poczta W, Średzińska J. (2007). Wyniki produkcyjno–ekonomiczne i finansowe indywidualnych gospodarstw rolnych według ich wielkości ekonomicznej (na przykładzie regionu FADN, Wielkopolska i Śląsk) (Production, economic and financial performance of individual farms grouped by economic size, as illustrated by the example of the Greater Poland and Silesia FADN region). *Problems of World Agriculture*, 2(17), 433–443.
- Ma Y., Zhang Z., Jang J., Qu J. (2019). Overcapacity investment and supervision Bunctuation: an evolutionary game approach. *Applied Economic Letters*.
- Bezat-Jarzębowska A., Rembisz W. (2015). Wprowadzenie do analizy inwestycji, produktywności, efektywności i zmian technicznych w rolnictwie (Introduction to the analysis of investments, productivity, efficiency and technical changes in agriculture). Institute of Agricultural and Food Economics, Warsaw, Poland.
- Zmyślona, J., Sadowski, A. (2019). Overinvestment in polish agriculture. Proceedings of the 2019 International Scientific Conference 'Economic Sciences for Agribusiness and Rural Economy'. No 3, Warsaw, 5–7 June 2019, 145–152.
- 33. Sharma R. (2019). Agricultural R&D Investment and Returns in India. *Suraj Punj Journal For Multidisciplinary Research*, 9(4), 39–46.
- Chen Z., Huffman W. E., Rozelle S. (2009). Farm technology and technical efficiency: Evidence from four regions in China. *China Economic Review*, 20(2), 153–161.
- 35. Feng Y., Zhang Y., Li S., Wang C., Yin X., Chu Q., et al. (2019). Sustainable options for reducing carbon inputs and improving the eco-efficiency of smallholder wheat-maize cropping systems in the Huan-ghuaihai Farming Region of China. *Journal of Cleaner Production*, 224, 118887.
- Poczta W., Sadowski A., Średzińska J. (2008). Rola gospodarstw wielkotowarowych w rolnictwie Unii Europejskiej (Role of large-scale agricultural holdings in the European Union's agriculture). Annuals of Agricultural Sciences. G series, agricultural economics, 95(1), 42–56.
- 37. Poczta W., Rowiński J. (2019). Struktura polskiego rolnictwa na tle Unii Europejskiej (Structure of Polish agriculture compared to the European Union). CeDeWu, Warsaw, Poland.
- 38. Sadowski A., Poczta W. (2007). Ocena skutków inwestycji wspieranych kredytem preferencyjnym dla gospodarstw rolnych (Assessing the effects of investments supported with soft loans for farms). Publishing House of the August Cieszkowski Agricultural University in Poznań, Poznań.
- Czubak W., Sadowski A. (2014). Wpływ modernizacji wspieranych funduszami UE na zmiany sytuacji majątkowej w gospodarstwach rolnych w Polsce (Impact of modernization projects co-funded by the EU on changes in the financial situation of Polish farms). *Journal of Agribusiness and Rural Development*, 2(32), 45–57.
- 40. Michalek J. (2012). Counterfactual Impact Evaluation of EU Rural Development Programmes—Propensity Score Matching Methodology Applied to Selected EU Member States. 1: A Micro-Level Approach; Publications Office of the European Union: Luxembourg.
- 41. Elhorst J. P. (1993). The estimation of investment equations at the farm level. *European Review of Agricultural Economics*, 20(2), 167–182.
- 42. Lim T. W., Shimazaki N., Godo Y., Lim Y. (2019). Comparison of the Impacts of Japan's Industrialization on the Coal Mining and Agriculture and Forestry Industries [in] *Coal Mining* Communities and Gentrification in Japan. Palgrave Macmillan, Singapore.
- Dries L., Swinnen J. F. M. (2010). The impact of interfirm on investment: Evidence from the Polish dairy sector. *Food Policy*, 35, 121–129.
- 44. Kardos M. (2014). The relevance of Foreign Direct Investment for sustainable development. Empirical evidence from European Union. *Procedia Economics and Finance*, 15, 349–1351.
- 45. Epaphra M., Mwakalasya A.H. (2017). Analysis of Foreign Direct Investment, Agricultural Sector and Economic Growth in Tanzania. *Modern Economy*, 8, 111–140.
- Mueller B., Mueller, Ch. (2016). The political economy of the Brazilian model of agricultural development Institutions versus sectoral policy. *The Quartely Review of Economics and Finance*, 62, 12–20.
- Panetto H., Lezoche M., Hernandez J., del Mar Eva M., Diaz A., Kacprzyk J. (2020). Agri-food 4.0: A survey of the supply chains and technologies for the future agriculture. *Computers in Industry*, 117, 103187.

- Szymczak J., Sadowski A. (2019), Technologia blockchain jako stymulanta zachowania bezpieczeństwa żywności w łańcuchu dostaw (Blockchain as a stimulator of food safety in the supply chain). Zagadnienia Doradztwa Rolniczego, 2 (96), 49–63.
- Klerx L., Rose D. (2020). Dealing with the game-changing technologies of Agriculture 4.0: How do we menage diversity and responsibility in food system transition pathways? *Global Food Security*, 24, 100347.
- Latruffe L., Balcombe K., Davidova S., Zawalinska K. (2004). Determinants of technical efficiency of crop and livestock farms in Poland. *Applied economics*, 36(12), 1255–1263.
- Tan S., Heerink N., Kruseman G., Qu F. (2008). Do fragmented landholdings have higher production costs? Evidence from rice farmers in Northeastern Jiangxiprovince, P.R. China. *China Economic Review*, 19(3), 347–358.
- Zhang X., Fan S. (2004). Public investment and regional inequality in rural" China. Agricultural Economics, 30, 89–100.
- 53. Schnabl G. (2019). China's Overinvestment and International Trade Conflicts. *China & World Economy*, 27(2), 1–27.
- 54. Knight J. and Ding S. (2010). Why does China invest so much? Asian Economic Papers, 9(3), 87–117.
- 55. Qin D., Song H. (2009). Sources of investment inefficiency: The case of fixed-asset investment in China. *Journal of Development Economics*, 90(1), 94–105.
- Lin Y.-C. (2017). Do Voluntary Clawback Adoptions Curb Overinvestment. Corporate Governance: An International Review, 25(4), 255–270.
- Wei X., Wang Ch., Guo Y. (2019). Does quasi-mandatory dividend rule restrain overinvestment? International Review of Economics and Finance, 63, 4–23.
- D"Mello R., Miranda M. (2010). Long-term and overinvestment agency problem. *Journal of Baking&Finance*, 34, 324–335.
- Nghia N.T., Khang T.L., Thanh N. C. (2019). The Moderation Effect of Debt and Dividend on the Overinvestment-Performance Relationship. [in:] Kreinovich V., Thach N., Trung N., Van Thanh D. (eds) Beyond Traditional Probabilistic Methods in Economics. *ECONVN. Studies in Computational Intelligence*, 809, 110–1120.
- Pellicani A. D., Kalatzis A. E. G. (2018). Ownership structure, overinvestent and underinvestment: Evidence from Brazil. Research in International Business and Finanse, 48, 475–482.
- Stiglitz J.E., Weiss A. (1981). Credit Rationing in Markets with Imperfect Information. *The American Economic Review*, 71, 393–410.
- **62.** Haramillo F., Schiantarelli F., Weiss A. (1996). Capital market imperfections before and after financial liberalization: An Euler equation approach to panel data for Ecuadorian firms. *Journal of Development Economics*, 51, 367–386.
- 63. Khemiri W., Noubbigh H. (2019). Does sub-Saharan Africa overinvest? Evidence from a panel of nonfinancial firms. *The Quarterly Review of Economics and Finance.*
- Lopez-de-Foronda O., Lopez-de-Silanes F., Lopez-Iturriaga F. J., Santamaria-Mariscal M. (2018). Overinvestment, laverage and financial system liquidity: A challenging approach. *Business Research Quarterly*, 22(2), 96–104.
- Badibanga T., Ulimwengu J. (2019). Optimal investment for agricultural growth and poverty reduction in the democratic republic of congo a two-sector economic growth model. *Journal Applied Economics*, 52 (2), 135–155.
- Negra Ch., Remans R., Attwood S., Jones S., Werneck F., Smith A. (2020). Sustainable agri-food investments require multi-sector co-development of decision tools. *Ecological Indicators*, 110, 105851.
- 67. Bański J. (2017). The consequences of changes of ownership for agricultural land use in Central European Countries following the collapse of Eastern Bloc. *Land Use Policy*, 66, 120–130.
- Bedoić R., Ćosić B, Duić N. (2019). Technical potential and geographic distribution of agricultural residues, co-products and by-products in the European Union. *Science of Total Environment*, 686, 568–579. https://doi.org/10.1016/j.scitotenv.2019.05.219 PMID: 31185404
- Chaplin H., Davidova S., Gorton M. (2004). Agricultural adjustment and the diversification of farm households and corporate farms in Central Europe. *Journal of Rural Studies*, 20, 61–77.
- Simoncini R., Ring I., Sandstrom C., Albert Ch., Kaymov U., Arlettaz R. (2019). Constraints and opportunities for mainstreaming biodiversity and ecosystem services EU's Common Agricultural Policy: Insights from the IPBES assessment for Europe and Central Asia. *Land Use Policy*, 88, 104099.
- Quiroga S., Suarez C., Fernandez-Haddad Z., Philippidis G. (2017). Levelling the playing field for European Union agriculture: Does the Common Agricultural Policy impact homogenously on farm productivity and efficiency? *Land Use-Policy*, 68, 179–188.

- Bartkowiak A., Bartkowiak P. (2017). Technical and Technological Progress in the Context of Sustainable Development of Agriculture in Poland. *Procedia Engineering*, 182, 66–75.
- Sadowski A., Wojtasiak J. (2019). Potencjał produkcyjny rolnictwa w krajach Unii Europejskiej (Production potential of agriculture in European Union countries). Zagadnienia Doradztwa Rolniczego 19(95), 5–18.
- 74. Czubak W. (2015). Nakłady inwestycyjne w rolnictwie polskim w kontekście wdrażania Wspólnej Polityki Rolnej Unii Europejskiej (Investment expenditure in the Polish agriculture in the context of implementing the European Union's Common Agricultural Policy) [in:] Czyżewski A. (eds.), Klepack i B. (eds.) Problemy rozwoju rolnictwa i gospodarki żywnościowej w pierwszej dekadzie członkostwa Polski w Unii Europejskiej (Problems of the development of agriculture and food economics encountered in the first decade of Poland's membership in the European Union), Polish Economic Society, Warsaw, Poland.
- 75. Poczta W., Sadowski A., Czubak W., Matyka M., Drygas M., Skórnicki H. (2017). Reforma Wspólnej Polityki Rolnej po 2020 roku (Reform of the Common Agricultural Policy post-2020), information brochure, conference materials, National Network of Rural Areas.
- 76. Góral J., Pilyavskyy A. (2018). The Common Agricultural Policy of the European Union–main challenges for a new budget [in:] The CAP and national priorities within the EU budget after 2020. Eds. Wigier M., Kowalski A., 72–84.
- 77. Ferto I., Bojnec S., Fogarasi J., Viira A. (2019). Agricultural soft budget constraints in new European Union member states. *Journal of Institutional Economics*, 16(1), 49–64.
- 78. Sapa A. (2001). Interwencjonizm w wymianie zagranicznej w sektorze rolno-żywnościowym wczoraj i dziś (Past and present of interventionism in international agri-food trade) [in] Today's problems of Polish agribusiness, eds. Czyżewski A., Publishing House of the University of Economics, Poznań, 52–63.
- 79. Stiglitz J.E. (1987). Some Theoretical Aspects of Agricultural Policies. The World Bank Research Observer, 2, no 1.
- Vasa L., Baranyai Z., Kovacs Z., & Szabo G. G. (2014). Drivers of Trust: Some Experiences from Hungarian Agricultural Cooperatives. Journal of International Food & Agribusiness Marketing, 26(4), 286– 297.
- Newman C., Briggeman B. C. (2016). Farmers' perceptions of building trust. International Food and Agribusiness Management Review, 19(1030-2016-83136), 57–76.
- Baranyai Z., Gyuricza C., Vasa L. (2012). Moral hazard problem and cooperation willingness: some experiences from Hungary. Actual Problems of Economics, (138), 301–310.
- 83. Lerman Z., Csaki C., Feder G. (2002). Land policies and evolving farm structures in transition countries. The World Bank, pp.49.
- Swinnen J. F., Gow H. R. (1999). Agricultural credit problems and policies during the transition to a market economy in Central and Eastern Europe. Food policy, 24(1), 21–47.
- Msaddak M., Ben-Nasr J., Zaibet L. (2020). Dynamics of Trust and Cooperation in the Dairy Value Chain: A Game Simulation Approach. Journal of International Food & Agribusiness Marketing, 1–21.
- Miller C., Ono T., Petruljeskov M. (2018). Agricultural investment funds for development: descriptive analysis and lessons learned from fund management, performance and private-public collaboration. Food and Agriculture Organization of the United Nations (FAO), pp. 56.
- **87.** Kovács Z. (2016). Moral hazard in producer organizations-some experiences of an empirical survey. Applied Studies in Agribusiness and Commerce, 10(4–5), 77–83.
- You J. (2014). Risk, under-investment in agricultural assets and dynamic asset poverty in rural China. China Economic Review, 29, 27–45.
- Guthrie G. (2006). Regulating infrastructure: The impact on risk and investment. Journal of Economic Literature, 44(4), 925–972.
- Sadowski A., Wojcieszak-Zbierska M. M., Beba P. (2021). Territorial differences in agricultural investments co-financed by the European Union in Poland. Land Use Policy, 100, 104934.